Author's response to reviews

Title: Correlates of objectively assessed physical activity and sedentary time in children: a cross-sectional study (The European Youth Heart Study)

Authors:

Andreas Nilsson (andreas.nilsson@oru.se)
Lars Bo Andersen (lars.bo.andersen@nih.no)
Yngvar Ommundsen (yngvar.ommundsen@nih.no)
Karsten Froberg (kfroberg@health.sdu.dk)
Luis B Sardinha (lsardinha@fmh.utl.pt)
Karin Piehl-Aulin (karin.piehl-aulin@oru.se)
Ulf Ekelund (ue202@medschl.cam.ac.uk)

Version: 5 Date: 3 September 2009

Author's response to reviews:

Dear Editor,

After proofreading the manuscript two minor typographical changes have been made. Number of 9-year-old girls presented in Table 1 and Table 3 had been reversed but is now correct. All other parts of the manuscript remain unchanged.

Sincerely yours,

Andreas Nilsson